Author's response to reviews

Title: Mononeuritis multiplex as the first presentation of refractory sarcoidosis responsive to etanercept

Authors:

Inês B Marques (i.marques@qmul.ac.uk)
Gavin Giovannoni (g.giovannoni@qmul.ac.uk)
Monica Marta (m.calado-marta@qmul.ac.uk)

Version: 2 Date: 1 September 2014

Author's response to reviews:

Dear editors,

We have revised the manuscript, namely the sections "patient consent", "author's contributions" and "competing interests" were added after the main text.